Author’s response to reviews

Title: High-performing physicians are more likely to participate in a research study: Findings from a Quality Improvement Study

Authors:

Simone Dahrouge (sdahrouge@bruyere.org)

Catherine Deri Armstrong (cdarmstrong@uottawa.ca)

William Hogg (whogg@bruyere.org)

Jatinderpreet Singh (jp.singh99@gmail.com)

Clare Liddy (cliddy@bruyere.org)

Version: 1 Date: 13 Mar 2019

Author’s response to reviews:

Please see document attached entitled "BMRM-D-18-00279 - Response to editor (2019-Mar-12)" which provides detailed responses to the comments in the decision letter.